Author's response to reviews

Title: A comparison of mental health policy and direct practice stakeholder perceptions of factors impacting evidence-based practice implementation using concept mapping

Authors:

Amy E Green (a4green@ucsd.edu)
Gregory A Aarons (gaarons@ucsd.edu)

Version: 4 Date: 3 June 2011

Author's response to reviews: see over
June 3, 2011

Denise O’ Conner, Ph.D.
Associate Editor
Implementation Science

Dear Dr. O'Connor:

We are pleased to submit our revised manuscript (MS: 1523426603422626) entitled “A Comparison of Policy and Direct Practice Stakeholder Perceptions of Criteria Affecting Evidence-Based Practice Implementation using Concept Mapping” for copy editing. We greatly appreciate your guidance and the reviewers’ comments and suggestions.

As requested we are submitting the attached documents as requested for copy editing:

1. A Microsoft word (.doc) copy of the manuscript (with revisions accepted and without changes tracked) as submitted during the second round of revision.

2. Two Powerpoint (.ppt) files representing Figure 1 and Figure 2. We believe .ppt to be the most appropriate format for these figures as it limits loss of digital information of the original figures while also allowing for additional editing. The figure legends are in the manuscript.

3. A Microsoft word (.doc) copy of the additional file which has undergone a final proofing by the authors.

Please contact me with any questions or concerns.

Sincerely,

Gregory A. Aarons, Ph.D.